Author’s response to reviews

Title: Refeeding syndrome in adults with celiac crisis : A case report

Authors:

Sonia hammami (sonia.hammami@fmm.rnu.tn)
Houda lazreg Aref (ibrahimhoudar@yahoo.fr)
Messouda Khalfa (Kh.messouda@yahoo.fr)
Ines Kochtali (Ines.kochtali@rnu.tn)
Mohamed Hammami (mohamed.hammami@fmm.rnu.tn)

Version: 1 Date: 04 Jan 2018

Author’s response to reviews:

Manuscript : JMCR-D-17-00571

Title: Refeeding syndrome in adults with celiac crisis : A case report

Authors: Sonia Hammami; Houda lazreg Aref; Messouda Khalfa; Ines Kochtali; Mohamed Hammami

Answers to Reviewer 1

Manuscript : JMCR-D-17-00571

Title: Refeeding syndrome in adults with celiac crisis : A case report

Authors: Sonia Hammami; Houda lazreg Aref; Messouda Khalfa; Ines Kochtali; Mohamed Hammami

4-d- …….Maybe authors could better describe on what morphological basis pathologists made diagnosis of celiac disorder
Answer: Serological tests and the result of intestinal biopsy were indicated in new version

The diagnosis of celiac disease was established by positive results of serological testing (tissue transglutaminase immunoglobulin, anti-gliadin, anti endomysium antibodies). The pathology report of intestinal biopsy revealed, villous atrophy with hyperplasia of the crypts and increased intraepithelial lymphocyte count. ....

Answers to Reviewer 2

Manuscript : JMCRR-D-17-00571

Title: Refeeding syndrome in adults with celiac crisis: A case report

Authors: Sonia Hammami; Houda lazreg Aref; Messouda Khalfa; Ines Kochtali; Mohamed Hammami

9- Please add a figure from the patient or an image technique

Answer: We add Fig 1 of the Anatomo-pathology exam